Author's response to reviews

Title: Epigenetic alterations of the keratin 13 gene in oral squamous cell carcinoma

Authors:

Kaori Naganuma (naganuma@college.fdcnet.ac.jp)
Mitsutoki Hatta (hatta@college.fdcnet.ac.jp)
Tetsuro Ikebe (ikb@college.fdcnet.ac.jp)
Jun Yamazaki (junyama@college.fdcnet.ac.jp)

Version: 4 Date: 12 December 2014

Author's response to reviews: see over
Dear Journal Editorial Office,

Thank you for your email of December 11, 2014, regarding our manuscript (MS: 1517799865138998) entitled “Epigenetic alterations of the keratin 13 gene in oral squamous cell carcinoma”. We have revised the manuscript according to the editorial suggestions.

- **Editorial request: “Please divide your “Discussion and Conclusions” section into two separate sections.”**

In accordance with the editorial request, we have divided “Discussion and Conclusions” section into “Discussion” and “Conclusions” sections. There is no change in the text of these sections.

We hope that the revised manuscript is now acceptable for publication in BMC Cancer.

Thank you for your consideration.

Sincerely yours,

Mitsutoki Hatta, D.D.S., Ph.D.
Department of Physiological Science and Molecular Biology, Fukuoka Dental College, Fukuoka 814-0193, Japan
Email: hatta@college.fdcnet.ac.jp